Title: Effects of an open-label pilot study with high-dose EPA/DHA concentrates on plasma phospholipids and behavior in children with attention deficit hyperactivity disorder

Authors:

  Paul J Sorgi (pjsorgi@aol.com)
  Edward M Hallowell (ehallowell@aol.com)
  Heather L Hutchins (hhutchins@eicosresearch.org)
  Barry Sears (bsears@drsears.com)

Version: 4 Date: 10 July 2007

Author’s response to reviews:

July 10, 2007

Dear Editors,

Thank you for the opportunity to make final revisions on our paper titled: Effects of an open-label pilot study with high-dose EPA/DHA concentrates on plasma phospholipids and behavior in children with attention deficit hyperactivity disorder. We have uploaded changes you requested for the figures and made final omissions and corrections in the text.

Thank you for your time.

Sincerely,

Heather Hutchins MS, RD